
P1 Kicker tank work 14243 Hillman / Wang
Relocation on RF Gun Kicker Thyratron box

Install Corrector Regulators in AH3's & Zone F 14244 Wang / Sprau AH3's Complete Zone F Started
Canted Undulator Install in S21 14242 Sprau Complete
Install Trims in S31, 24 & 25 14246 McNamara Installed/Test @ Startup
Remove Trims from 14, 26 & 27 14247 McNamara Complete
Replace gespac in S22 cab 2 14648 McNamara Complete
Tie in ACIS chain C into the MPEOB string 14522 McNamara / Hogrefe Complete/Needs Tested

Support IS1 replacement 14521 McNamara
Reconnecting Today (Waiting on Bake 

out to finish)
General Septum Maintenance 14520
General Kicker Maintenance 14520
Investigate EMW overcurrent trips 14523 McNamara / Deriy
Operator Training Sprau Complete
Add Cooling fans to SR Converter Cabinets 14245 Markovich Complete
General PM

PS SR Tasks

PS Injector Tasks
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Duration  Assigned Start Date
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Linac / Par fiber replacement Hogrefe
Install Clock synchronizer from Linac/Par to Storage Ring Forrestal / Hogrefe
Verify operation of the HET Double Stop after cylinders are changed by the vacuum group Forrestal / Hogrefe Budz
Verify operation of the LEUTL Single Stop after the cylinder is changed by the vacuum group Forrestal / Hogrefe Budz
Modify top shield & A005 Storage Ring ACIS rack wiring Forrestal / Hogrefe Budz
Map Zone F Area RIO links Forrestal / Hogrefe Budz
Install Chain C connections from last BSS in each zone to the PS MPEOB's for that zone Hogrefe / McNamara
Identify 420 cabling for contractors to install for wave guide shutters Forrestal / Markovich
Validate Storage Ring Forrestal / Hogrefe Budz
Exercise all installed front end shutters for sectors 1 - 34 (part of Storage Ring validation) Forrestal / Hogrefe Budz
Verify operation of S39 viewscreen after work by B. Berg Forrestal / Hogrefe
Verify operation of Radiation Monitor @ 30ID after relocation Forrestal / Hogrefe Budz
Verify operation of Radiation Monitor @ 29ID after installation Forrestal / Hogrefe Budz

17BM Reterminate cable ends after front end valve is replaced 14502 Berndt 1/2 d
33ID Pull and terminate cables for Beamline isolation Valve 14457 Berndt 1 d
16 Upgrade V2 FEEPS to V2.1 14571 Behrndt Kudirka 1/14/2004 1d
22 Upgrade V2 FEEPS to V2.1 14572 Behrndt Kudirka 1/15/2004 1d
21 FEEPS Install, Cabling 14568 Behrndt Gomez, Oakley 1/8/2004 1d

FEEPS Tasks

ACIS Tasks

Comments

Dec / Jan Shutdown 12/23/03 - 01/22/04

Responsible Eng.  Estimated 
Duration  

Assigned Techician 
ResourcesWORK Items WRQ # Assigned Start Date

Phil Mcnamara Dec 03 Shutdown Work List.xls 1/16/2004



9ID Validation Dawson Gomez 5-Jan 3d Complete
11ID Validation Hawkins 5-Jan 2d Complete
16ID Validation Servino Budz 7-Jan 3d Complete
14BM Validation Roberts 8-Jan 2d Complete
19BM Validation Nguyen Kudirka 12-Jan 2d Complete
19ID Validation Gomez 12-Jan 2d Complete
22ID Validation Servino Budz 14-Jan 2d Complete
13BM Validation Nguyen 19-Jan 2d
31ID Validation Emmerson Kudirka 19-Jan 2d
20BM Validation Roberts 21-Jan 2d
17BM Validation Servino Budz 27-Jan 3d
8BM Validation Gomez 9-Feb 2d

All Inspect FE Shutters for radiation damage Freidman Gomez 12/29/2003 2d Complete
4ID Relocate BLEPS rack #2 to roof 13834 Freidman Kudirka 12/29/2003 5d Complete

17BM BLEPS-to-PSS interface 14468 Nguyen Roberts 1/7/2004 1d Complete
17BM Install Remote Shutter Interface 14237 Freidman Roberts 1/7/2004 1d Complete
17BM Integrate P6IS 14234 Freidman Roberts 1/7/2004 2d Complete
ALL Replace WDT relays 14466 Nguyen Gomez 12/23/2003 1/2d Complete

13BM Integrate new Shutter SH1-C 14507 Freidman Roberts 1/5/2004 2d Complete
8BM Install Remote Shutter Interface 14453 Freidman Roberts 1/15/2004 1d Complete
8BM BLEPS-to-PSS interface 14454 Nguyen Roberts 1/15/2004 1d Complete
8BM Assist beamline with wiring issues McNamara Oakley 1/12/2004 3d Complete
5ID Relocate DIW units 14527 Dawson Gomez 1/14/2004 2d On Going
7ID Relocate DIW units 14527 Dawson Gomez 1/14/2004 2d On Going
8ID Relocate DIW units 14527 Dawson Gomez 1/14/2004 2d On Going
ALL ACIS - PSS Validations 14528 Freidman On Going
ALL Exercise all FE shutters 14529 Freidman Complete
21ID Remove old cables and pull in new one between FERDP and device control box Hawkins Gomez, Oakley 1/8/2004 1d Complete
24ID Continue on with construction Hawkins Background Task

Repair / Replace cables damaged by radiation Freidman Complete

WORK ItemsBeamline
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